Author's response to reviews

Title: Temporal changes in haematocrit following artemisinin-based combination treatments of uncomplicated falciparum malaria in children

Authors:

Akintunde Sowunmi (akinsowunmi@hotmail.com)
Kazeem Akano (heycarnow@yahoo.com)
Adejumoke I Ayede (idayede@yahoo.com)
Godwin Ntadom (ntadomg@yahoo.com)
Bayo Fatunmbi (bayofat@gmail.com)
Temitope Aderoyeje (tope_forever@yahoo.com)
Elsie O Adewoye (elolade@yahoo.com)

Version: 2 Date: 11 February 2015

Author's response to reviews:

1. Line Numbering: line and page numbers have been included in the manuscript. In addition, page breaks in the manuscript have been removed.

2. List of abbreviations: List of abbreviations used in the text has been included in the manuscript

3. All authors email addresses are included on the title page.